Author's response to reviews

Title: Low carbohydrate diets in family practice: what can we learn from an internet-based support group

Authors:

   Richard D Feinman (rfeinman@downstate.edu)
   Mary C Vernon (mvernonmd@yahoo.com)
   Eric C Westman (ewestman@duke.edu)

Version: 2 Date: 22 September 2006

Author's response to reviews:

Formatting changes according to production requests